Author’s response to reviews

Title: Does medical training in Thailand prepare doctors for work in community hospitals? An analysis of critical incidents

Authors:

Dumrongrat Lertrattananon (dumrongrat.ler@mahidol.edu)
Wirun Limsawart (lwirun@gmail.com)
Alan Dellow (alan.dellow@btinternet.com)
Helen Pugsley (pugsleyh3@cardiff.ac.uk)

Version: 1 Date: 25 Jun 2019

Author’s response to reviews:

Reviewer1

1. I have added a paragraph in the background regarding medical curricula in Thailand.

2. & 5. I have added some references for examples of utilization of critical incident technique in health sector and a previous use of this technique to explore doctor's perception of preparedness for practice. Also, I discussed about this in the discussion.

3. I have mentioned about the recall bias in the discussion.

4. I have revised table 1 and added table 2. The analysis did not show differences in preparedness for practice regarding size or region of hospitals and the undergraduate tracks.

6. More specific recommendations were added.

Reviewer2

1. I have added more results and specified all numbers.

2. I have cut these sentences.